Author’s response to reviews

Title: The relationship between prevalence and severity of lower urinary tract symptoms (LUTS), and body mass index and mid-abdominal circumference in men in a resource-poor community in southeast Nigeria: a cross-sectional survey

Authors:

Ikenna Nnabugwu (ikenna.nnabugwu@unn.edu.ng)

Fredrick Ugwumba (fredrick.ugwumba@unn.edu.ng)

Emeka Udeh (ihechiudeh@yahoo.com)

Solomon Anyimba (sanyimba@yahoo.com)

Louis Okolie (louisbenokolie@gmail.com)

Version: 2 Date: 25 Jan 2019

Author’s response to reviews:

Covering Letter to 2nd Review

1. Consent to participate: This has been clarified in the consent section. In last but one line of page 5, it is stated.

2. Figures: Figures 1 and 2 have been removed from the main manuscript and submitted as separate files. The “figure legends” section have been created after the References.

3. Tables: Table 1 has been moved to the end of the main manuscript.

4. Conclusions: The “conclusions” section has been created.

5. Abstract: The corrections have been made.

6. Clean copy: Track changes, line pages have been removed and a clean copy of manuscript, abstract, figure 1 and figure 2 submitted. The older versions have been done away with.

Thank you.